## **Author Response to Reviewer Comments**

Date: [Insert Date]

[Your Name]

[Your Institution]

[Your Email]

Editorial Office [Journal Name]

## Dear [Editor's Name],

Thank you for the opportunity to revise our manuscript titled "[Manuscript Title]" (Manuscript ID: [ID]). We also appreciate the constructive feedback from the reviewers. Below, we provide detailed responses to each comment.

## **Responses to Reviewer 1:**

**Comment 1:** [Reviewer comment]

**Response:** We appreciate this insight and have made the following changes: [Details of changes made].

**Comment 2:** [Reviewer comment]

**Response:** Thank you for this suggestion. We have [Details of changes made].

## **Responses to Reviewer 2:**

**Comment 1:** [Reviewer comment]

**Response:** We agree with the comment and have revised the manuscript accordingly: [Details of changes made].

**Comment 2:** [Reviewer comment]

**Response:** In response, we [Details of changes made].

We have carefully revised the manuscript to address all the comments and believe that these changes have strengthened our work. Thank you again for your consideration.

Sincerely, [Your Name]